Author’s response to reviews

Title: Ontological security and connectivity provided by pets: A study in the self-management of the everyday lives of people diagnosed with a long-term mental health condition

Authors:

Helen Brooks (Helen.Brooks@manchester.ac.uk)
Kelly Rushton (kelly.rushton@manchester.ac.uk)
Sandra Walker (S.C.Walker@soton.ac.uk)
Karina Lovell (karina lovell@manchester.ac.uk)
Anne Rogers (A.E.Rogers@soton.ac.uk)

Version: 3 Date: 01 Nov 2016

Author’s response to reviews:

You state in your 'consent for publication' declaration that "All participants provided consent for the recording and publishing of anonymised data."

During our final editorial assessment it has come to our attention that two of the quotes provided within your manuscript contain information which is potentially identifiable. The quote on page 12-13 of your manuscript refers to 'Diane'. In addition, the quote on page 23 refers to 'Karen’s nan'. Please either remove these quotes or modify them to remove the identifiable information.

I removed the names Diane and Karen and replaced with [friend]. I hope this is satisfactory. Should you require any further information, please do not hesitate to contact me.

Kind regards,

Helen Brooks
Research Fellow